Author's response to reviews

Title: Indapamide-induced transient myopia with supraciliary effusion: case report

Authors:

Mihaly Vegh (contactology52@freemail.hu)
Andras Hari-Kovacs (harikovacs@gmail.com)
Kata Rez (rez_kata@se-etk.hu)
Beata Tapaszto (beatatapaszt@gmail.com)
Agnes Szabó (szabo.agnes@med.u-szeged.hu)
Andrea Facsko (aafacsko@gmail.com)

Version: 2 Date: 4 June 2013

Author's response to reviews: see over
Dear Editors,

You find enclosed a manuscript titled “Rare case of indapamide-induced transient myopia with supraciliary effusion.” by M Vegh and Andras Hari-Kovacs, as shared first authors, and co-writers which is submitted for a consideration to be published by BMC Ophthalmology for the following reasons:

1; It reports on a very rare ocular adverse effect of a widely used antihypertensive medication.
2; To our best knowledge, this would be the third publication about indapamide-induced myopia, and the first study to employ ultrasound biomicroscopy for describing its morphological characteristics.
3; Our study is the first to reveal that both ciliary muscle contraction and ciliary body oedema may play role in the patomechanism.

The material has not been submitted for publication to any other journals. The authors are ready to sign a “Transfer of Copyright” agreement before publishing the article and agree that if it will have published in the BMC Ophthalmology will not be reprinted elsewhere in any language in the same form without the consent of the publisher.

Your kind work concerning the manuscript is gratefully appreciated. Should you have any query please do not hesitate to contact me.

Yours sincerely,

Mihaly Vegh MD, PhD